Author's response to reviews

Title: Influence of ultra-low dose Aprotinin on thoracic surgical operations: a prospective randomized trial

Authors:

Efstratios Apostolakis (stratisapostolakis@yahoo.gr)
Nikolaos Panagopoulos (ekoletsis@upatras.gr)
Efstratios Koletsis (ekoletsis@hotmail.com)
James Crockett (fccp_perfusionist@yahoo.co.uk)
Helen Stamou-Kouki (elenistamou21@hotmail.com)
Efrosini Sourgiadaki (ekoletsis@upatras.gr)
Kriton Filos (ekoletsis@upatras.gr)
Dimitrios Dougenis (ddougenis@med.upatras.gr)

Version: 3 Date: 11 March 2008

Author's response to reviews: see over
Dear editor

Re: Influence of ultra-low dose Aprotinin on thoracic surgical operations: a prospective randomized trial

No: MS: 1726199408147533

Thank you for editing this paper for publication. A new editing procedure took place and all the amendments concerning the tables have been performed in the manuscript.

Thank you in advance

Efstratios Koletsis MD, PhD
Senior Lecturer
University of Patras